\begin{table}%t8
\caption{{\it EW}s for neutron capture element lines in the Ba-star~\#6007741 in \object{NGC~6441}.}\label{t:ncaptureew}
\small%\centerline
{
\begin{tabular}{lccccc}
\hline \hline
Element & Wavelength & E.P. & $\log{gf}$ &Ref. & {\it EW} \\
      & (\AA)   & (eV) &       &               &(m\AA) \\
\hline
Y~I   & 6222.58 & 0.00 & --1.70 & 1 &   78.0 \\
Y~II  & 5728.89 & 1.84 & --1.12 & 1 &   50.0 \\
Zr~I  & 6127.46 & 0.15 & --1.06 & 2 &  145.0 \\
Zr~I  & 6134.57 & 0.00 & --1.28 & 2 &  134.0 \\
Zr~I  & 6140.46 & 0.52 & --1.41 & 2 &  100.0 \\
Zr~I  & 6313.03 & 1.58 &  0.27 & 2 &   86.0 \\
Ba~II & 6141.75 & 0.70 &  0.00 & 3 &  382.1 \\
La~II & 5769.06 & 1.25 & --0.69 & 4 &   88.0 \\
La~II & 5797.57 & 0.24 & --1.36 & 4 &  107.0 \\
La~II & 6390.48 & 0.32 & --1.41 & 4 &   90.0 \\
Ce~II & 5613.69 & 1.42 & --0.47 & 5 &   35.0 \\
Nd~II & 5740.86 & 1.16 & --0.53 & 6 &   54.0 \\
Nd~II & 5804.00 & 0.74 & --0.53 & 6 &   69.0 \\
Nd~II & 5811.57 & 0.86 & --0.86 & 6 &   50.0 \\
Nd~II & 5825.86 & 1.08 & --0.66 & 6 &   52.0 \\
Nd~II & 6382.06 & 1.44 & --0.75 & 6 &   46.0 \\
Eu~II & 6313.03 & 1.28 & --1.02 & 7 &$<$20.0 \\
\hline
\end{tabular}}
\medskip
1. Hannaford et~al. (\cite{Hann82});
2. Bi\'emont et~al. (\cite{Biem81});
3. Holweger \& M\"uller (\cite{Holw74});
4. Lawler et~al. (\cite{Lawl01b});
5. Bi\'emont et~al. (\cite{Biem05});
6.~den~Hartog et~al. (\cite{denh03});
7. Lawler et~al. (\cite{Lawl01a}).
\end{table}